Author's response to reviews

Title: Homocysteine and reactive oxygen species in metabolic syndrome, type 2 diabetes mellitus, and atheroscleropathy: The pleiotropic effects of folate supplementation

Authors:

Dr Melvin R Hayden M.D. (mrh29@usmo.com)
Suresh C Tyagi Ph.D. (s0tyag01@louisville.edu)

Version: 2 Date: 3 Mar 2004

Table 4 was condensed into a continuous table with no interruptions such as "table 3 continued or duplication of column headings. Additionally, the list of abbreviations was inserted into the text in the proper location. Now there exists only one table 4 and there was no need to renumber the tables or change the order of their appearance in the text. The figures have been reviewed and are ready for proofing. The text has been reviewed and is ready for proofing and publications. Thank you for this recommendations as I am certain it will improve the appearance of the manuscript.

Sincerely,

M.R. Hayden, M.D. corresponding author